Author's response to reviews

Title: Trends in sexually transmitted infections in the Netherlands, combining surveillance data from general practices and sexually transmitted infection centers

Authors:

Ingrid VF van den Broek (ingrid.van.den.broek@rivm.nl)
Robert A Verheij (R.Verheij@nivel.nl)
Christel E van Dijk (C.vanDijk@nivel.nl)
Femke DH Koedijk (femke.koedijk@rivm.nl)
Marianne AB van der Sande (Marianne.van.der.Sande@rivm.nl)
Jan EAM van Bergen (JvanBergen@soaaid.nl)

Version: 5 Date: 17 May 2010

Author's response to reviews:

Dear Editor,

My apologies for being late, I had a short break from work and did not check my e-mail.

I have made changes in the title page: the addresses were given as 6 separate adresses, whereas the authors come from 3 different institutes only, so it is important that this is changed. Also 1 email address was not correct. I changed the abbreviations for the author names to be in line with the section ‘author contributions’.

thank you, best regards,

Ingrid van den broek